#### Número 544

# RODOLFO CERMEÑO,<sup>\*</sup> MARÍA ROA GARCÍA<sup>\*\*</sup> AND CLAUDIO GONZÁLEZ-VEGA<sup>\*\*\*</sup>

# Financial Development and Growth Volatility: Time Series Evidence For Mexico and The United States

#### Importante

Los Documentos de Trabajo del CIDE son una herramienta para fomentar la discusión entre las comunidades académicas. A partir de la difusión, en este formato, de los avances de investigación se busca que los autores puedan recibir comentarios y retroalimentación de sus pares nacionales e internacionales en un estado aún temprano de la investigación.

De acuerdo con esta práctica internacional congruente con el trabajo académico contemporáneo, muchos de estos documentos buscan convertirse posteriormente en una publicación formal, como libro, capítulo de libro o artículo en revista especializada.

#### DICIEMBRE 2012



www.cide.edu

<sup>\*</sup> División de Economía, Centro de Investigación y Docencia Económicas, CIDE, México.

<sup>\*\*</sup> Centro de Estudios Monetarios Latinoamericanos, CEMLA, México.

<sup>\*\*\*</sup> Department of Agricultural, Environmental and Development Economics, The Ohio State University, Columbus, Ohio.

D.R. © 2012, Centro de Investigación y Docencia Económicas A.C. Carretera México Toluca 3655, Col. Lomas de Santa Fe, 01210, Álvaro Obregón, México DF, México. www.cide.edu

Dirección de Publicaciones publicaciones@cide.edu Tel. 5081 4003

••••••••••••••••••••••

## Abstract

This paper explores the influences of financial deepening on growth and its volatility. Following a review of the theoretical literature that has attempted to explain these relationships, the paper presents time series evidence for the cases of Mexico and the US. The results suggest that, in the case of the US, financial deepening has been related to the rate of real output growth but that finance has not shown a significant relationship with output volatility. In the Mexican case, financial deepening has reduced the volatility of growth which, in turn, has induced higher output growth rates. Further, higher US growth rates have resulted in higher and less volatile growth rates in the Mexican economy.

Keywords: Financial development, monetary and credit deepening, growth, volatility, VAR models, GARCH models.

JEL Classification: C22, C32, F43, O40.

#### Resumen

En este trabajo se explora la influencia de la profundización financiera sobre el crecimiento del producto real y su volatilidad. Después de hacer una revisión de la literatura teórica que intenta explicar estas relaciones, se ofrece evidencia de series de tiempo para los casos de México y Estados Unidos. Los resultados sugieren que en el caso de Estados Unidos, la profundización financiera se relaciona con la tasa de crecimiento del producto real, pero no con su volatilidad. En el caso de México, la profundización financiera ha reducido la volatilidad del crecimiento lo cual, a su vez, ha tenido un efecto positivo sobre la tasa de crecimiento. Adicionalmente, se encuentra que mayores tasas de crecimiento del producto en Estados Unidos han resultado no solamente en mayores tasas de crecimiento sino también en una menor volatilidad del crecimiento en México.

Palabras clave: Desarrollo financiero, profundización monetaria y crediticia, crecimiento, volatilidad, modelos VAR, modelos GARCH.

Clasificación JEL: C22, C32, F43, O40.

# Introduction

For at least a couple of centuries, the influence of financial development on economic growth has attracted vigorous debate among economists. Despite numerous approaches —within the current consensus— on what circumstances may actually produce these effects, there is growing empirical evidence that financial variables have significantly influenced the rate of economic growth.

On the one hand, the theoretical literature has identified alternative mechanisms through which the performance of the financial system influences the fundamental determinants of economic growth. In particular, the accumulation of physical and human capital and technological innovation are spurred by the role of the financial sector in mobilizing and pooling savings mostly from households (surplus units) and reallocating this purchasing power to investment projects with high marginal rates of return (deficit units) as well as its roles in improving the stock of information about investment opportunities and firm performance, the monitoring of managers and exercise of corporate control, the pooling, exchanging, diversifying and mitigating of idiosyncratic and systemic risk and, in general, its role in.

On the other hand, the empirical literature suggests that a better performance of the financial system leads to higher output growth rates, although the specific channels for these effects are not fully specified (Beck, Levine and Loayza, 2000). Further, both the theoretical and the empirical contributions recognize and discuss issues about reverse causality; indeed, economic growth also influences financial development.

In turn, there is a literature —albeit not as developed— that examines the influence of financial deepening on the volatility of economic growth. Here as well, theoretical contributions have identified mechanisms through which finance may influence volatility. In particular, by diversifying production risks, smoothing responses to liquidity shocks, contributing to the mobilization of savings —as precautionary reserves— and improving the stock of information, the efficient performance of the financial sector may diminish the volatility of output growth. Empirical contributions seem to support the theoretical predictions in this case as well. The more recent papers emphasize that the rate and the volatility of economic growth are influenced, not just by the level of financial development, but also by the structure of the financial sector and, in particular, by the evolution over time of the banking system.

The objective of this paper is to assess the influence of financial deepening on the rate and volatility of output growth in two quite different but closely interconnected economies: Mexico and the United States, using time series methods. First, unit root tests are carried out to determine if the variables do exhibit stochastic trends. Next a VAR approach is used in order to examine Granger-causality between finance and growth. The relationship

between measures of financial development and growth volatility is then investigated using GARCH models. In all cases, diagnostic checks, particularly autocorrelation tests, are implemented to assure that the estimated models are well specified.

The results suggest that, in the case of the United States, financial deepening has been positively related to the rate of economic growth but that it has not been significantly related to the volatility of growth. In turn, in the case of Mexico, the results show that financial deepening has had a positive impact on economic growth through the channel of reducing the volatility of growth. When financial development is approximated by a money deepening variable, the results are stronger. This suggests that the expansion of banking has had a positive effect on economic growth in Mexico. When, instead, the proxy used is domestic credit, the results are ambiguous. These results suggest a potentially negative effect of an ample use of domestic credit to fund government budgets rather than private investment. Finally, this study finds that higher US growth rates have resulted not only in higher growth rates in the Mexican economy, a fact that is well known, but also in less volatile growth. To the best of our knowledge, this last finding, on the effect of US growth on growth volatility in Mexico, is novel in that it explicitly identifies a volatility channel and has important implications for understanding the links between these economies.

The remainder of the paper proceeds as follows. Section 1 reviews the theoretical and empirical literature. Section 2 gives an overview of the historical and current situation of the financial system in Mexico and in the United States. Section 3 describes the time series methodology used in the study. Section 4 presents and discusses the empirical results. The main conclusions are summarized in Section 5.

# 1. Theoretical Contributions

## 1.1. Financial Development and Economic Growth

Interest in the relationship between financial institutions and economic growth is not new. Earlier, when exploring the role of institutions, Hamilton (1791) and Bagehot (1873) and then Schumpeter (1934) and Hicks (1969) had looked into this relationship. Attention to the connection between finance and growth increased in the second-half of the last century (Gurley and Shaw, 1955 and 1960; Cameron et al., 1967; Goldsmith, 1969; McKinnon, 1973 and 1976; Shaw, 1973).<sup>1</sup> These authors identified transaction and information

<sup>&</sup>lt;sup>1</sup> As Levine (1997) highlights, the pioneers analyzed the role of finance in economic growth with models that formalized the financial sector solely in terms of money and introduced a distinction between the financial and real sectors of the economy. Nevertheless, as these more recent contributions have highlighted, the financial sector is "real". Based on their approach, Fry (1988 and 1995) examines several models of growth with money, including

costs (in the exchange of goods and securities and in enforcing contracts) as the trigger for the emergence of financial markets and intermediaries. In their view, financial tools seek to mitigate the economic consequences of frictions, in order to enlarge market size and arrive at more efficient allocations of resources, which increase factor productivity and accelerate economic growth.

Others, however, have questioned the role of finance in economic growth and have claimed that financial deepening is a consequence, not a cause, of economic growth (Robinson, 1952; Lucas, 1988). In particular, the traditional literature on economic growth (Romer, 1986 and 1990; Lucas, 1988; Aghion and Howitt, 1992) ignored the role of financial institutions as an engine of economic growth. The main reason was that these endogenous growth models were developed in the general-equilibrium framework of Arrow-Debreu and did not allow for frictions and imperfections. Thus, during the blossoming of economic growth models, the role of financial development was overlooked.

Towards the end of the century, however, interest in identifying a positive influence of financial development on economic growth resurfaced. In contrast to the earlier contributions, which focused on the role of money, a growing new literature has developed models that explicitly specify the relationship between financial intermediation and growth and has attempted to empirically verify their implications. These new contributions share a common thread with the pioneers: financial institutions emerge to mitigate the problems that result from the existence of transaction and information costs (which, in turn, induce adverse selection and moral hazard problems) and to facilitate responses to liquidity shocks.

In attempting to address these challenges, the financial sector influences the ultimate determinants of growth, as established by the traditional literature on economic growth: the accumulation of physical and human capital (Lucas, 1988; Romer, 1986) and technological change (Romer, 1990; Grossman and Helpman, 1991; Aghion and Howitt, 1992). Thus, starting from endogenous growth models, these contributions eliminate the assumption of perfect financial markets and thereby manage to incorporate the influence of changes in the efficiency of financial markets on economic growth.<sup>2</sup> To accomplish this, they generally consider the existence of credit rationing and take this outcome as an indicator of the level of financial development in a given country.

A full review of the theoretical discussion can be found in Levine (2004). The author classifies contributions according to how the different functions of the financial system influence the determinants of economic growth. In

Kapur (1976), Galbis (1977), and Mathieson (1980) as well as the contributions of Spellman and of Gonzalez-Vega, included in McKinnon (1976).

 $<sup>^2</sup>$  Classical models of exogenous growth do not offer an appropriate framework, to the extent to which financial variables influence the levels rather than the rates of growth of output (Pagano, 1993).

particular, the financial system generates information on potential investment projects, allocates funds to the most profitable ones, and monitors their performance, thereby improving corporate governance. The financial sector also allows the diversification and management of risk, mobilizes and pools savings and facilitates the exchange of goods and services, assets and factors of production. Levine concludes that, despite the diversity of approaches, there is strong evidence that financial variables have a significant impact on economic growth. Some key contributions are summarized next.

Some of the earlier contributions used the conceptual framework of the *AK* model and showed how financial development influences the accumulation of physical capital (Greenwood and Jovanovic, 1990; Bencivenga and Smith, 1991 and 1993; Levine, 1991 and 1992), either by increasing the savings rate or by reallocating savings to technologies with a higher productivity of capital.

The more recent contributions belong to two sets. On the one hand, there are papers that examine the contribution of the financial system in the process of human capital formation. Access to credit to fund education is constrained by asymmetric information problems between borrowers and lenders and because human capital cannot be pledged as collateral (Galor and Zeira, 1993; Buiter and Kletzer, 1995; De Gregorio, 1996; Jacoby and Skoufias, 1997; De Gregorio and Kim, 2000).

On the other hand, some contributions follow the evolution of economic growth theories and, in recovering the ideas of Schumpeter, focus on showing how the development of the financial sector promotes research and development (R&D) at the firm level, accelerating technological change and growth (King and Levine, 1993; De la Fuente and Marín, 1996; Kower, 2002; Morales, 2003; Acemoglu, Aghion and Zilibotti, 2002; Aghion et al., 2004). Recently, Levine (2009) develops an endogenous growth model in which not only entrepreneurs undertake innovation, but also the financial sector searches for new processes to evaluate entrepreneurs. The conclusion is that technological innovation by firms and economic growth would cease unless the financial sector also innovates.

There is as well an ambitious collection of empirical contributions in the literature. Levine (2004) offers, again, a complete review. These contributions use different techniques and methods: growth regressions for a cross-section of countries (Goldsmith, 1969; King and Levine, 1993; Levine and Zervos, 1998; Levine, Loayza and Beck, 2000; La Porta et al., 1999), time series analysis (Jung, 1986; Demetriades and Hussein, 1996; Arestis et al., 2000; Ang and McKibbin, 2007) and panel techniques (Levine, Loayza and Beck, 2000; Beck, Levine and Loayza, 2000; Loayza and Ranciere, 2002). There are also detailed country studies and recent approaches, based on microeconomics, explore these issues at the industry or firm level (Rajan and Zingales, 1998; Ahlin and Jiang, 2005; Aghion, Fally and Scarpeta, 2006). Using a model to verify the costs levied by governments on financial

intermediation, Greenwood et al. (2010) show, for the United States during 1974 to 2004 first, and then with international data, that most of these countries could increase their output if they had a more efficient financial sector.

In general, all these contributions suggest that countries with a better performing financial sector enjoy higher rates of growth. A better performance of the banking sector would reduce the financial constraints that limit the expansion of firms and industries and this would be one of the channels of influence of the financial sector on economic growth. Levine (2004) also highlights that a common problem with these exercises is that the proxy variables used as indicators of financial development do not directly reflect the functions performed by the financial sector, in the ways claimed by the theoretical contributions.

Several contributions combine the influence of finance with other determinants of growth. Using data from Chinese provinces, Hassan et al. (2009) show that the development of financial institutions, legal regime, property rights, and political pluralism are associated with a stronger growth performance. Using data from 100 countries, Giulianoa and Ruiz-Arranz (2009) find that remittances encourage economic growth in countries with a low level of financial development, as they offer an alternative for the funding of investment projects and in overcoming liquidity constraints. Masten et al. (2008) relate financial development to international agreements. In the case of the European Union, they find that financial development and integration among the countries in the Union has had positive effects on growth. In particular, the effect of financial developming has been greater in the least developed countries, while the integration effect has been greater in the countries with the greatest level of financial development.

A potential challenge for the empirical analysis is reverse causality; the level of economic activity and technological change may, in turn, influence financial development. On the one hand, innovations in telecommunications and data management have reduced transaction costs and have encouraged the development of new financial products (Merton, 1992; Gup, 2003). On the other hand, economic development encourages savers and investors to channel resources to the financial system (Greenwood and Jovanovic, 1990). As Levine (2004) recognizes, however, despite the importance of an analysis of how the structure of the financial system influences and is influenced by economic growth, the corresponding literature has been considerably less developed.

An exception is Fung (2009), who empirically explores the potential convergence of financial development and economic growth and the interaction between them. The results confirm the existence of conditional convergence. Middle-income and high-income countries tend to converge, not only with respect to their per capita GDP, but also with respect to financial

development. The relationship and reciprocal interaction between financial development and economic growth is particularly strong in the early stages of development, and it tends to dilute once countries settle on a steady-state path. Countries with low incomes but with a healthy financial development catch up with middle income countries, while those countries that lack a well-performing financial system are caught in a poverty trap. The empirical section of this paper takes a closer look at the issue of causality and it analyzes the direction of causality between these two processes in the countries under study.

## 1.2. Financial development and growth volatility

The literature on financial development and economic growth has focused on explaining the influence of the performance of the financial sector on the positive trend of output growth and until recently —mostly now because of the global financial crisis— it had paid scant attention to its impact on output fluctuations. A somewhat less-developed literature thus examines how financial development influences the volatility of economic growth. The existence of persistent oscillations around the trend of output growth is, however, one of the fundamental stylized facts of economic growth. Further, the most recent literature discusses how the current crisis may have been influenced by the extent to which financial intermediaries have been one of the propagating mechanisms, a role that in turn depends on their level of development.

Let's look first at the literature that explores the effects of volatility itself on the rate of economic growth. While the empirical contributions (Ramey and Ramey, 1995; Blattman et al., 2004; Koren and Tenreyro, 2004; Aghion et al., 2004) find a negative correlation between volatility and economic growth, theoretical treatments disagree with these results, claiming that volatility may be either positively or negatively correlated with growth. Jones et al. (2000) conclude that the sign of the relationship between volatility reduces the risk-adjusted returns on investment, thereby discouraging investment and growth. On the other hand, greater volatility increases precautionary savings and, through this channel, it accelerates economic growth. Which one of the two effects dominates depends on the value of the elasticity of intertemporal substitution.

In the literature on financial development and volatility there are three theoretical strands, which analyze the mechanisms through which the financial system's performance may influence the volatility of growth. They are based on any one of the functions performed by financial intermediaries (Levine, 1997 and 2004). By diversifying portfolios, facilitating the management of production risk and coping with liquidity shocks, mobilizing the precautionary savings of households, and generating information about the risks and returns of alternative investments (in order to allocate capital more efficiently), the financial sector may diminish the volatility of economic growth.

Portfolio theory is one of these approaches. From this perspective, financial development implies the creation of different instruments for risk diversification, which facilitate investment in more risky but more profitable assets. Diversification not only encourages growth, but it reduces uncertainty (Greenwood and Jovanovic, 1990; King and Levine, 1991; Levine, 1991; Saint-Paul, 1992; Devereux and Smith, 1994; Obstfeld, 1994; Acemoglu and Ziliboti, 1997).

The general framework for these contributions is as follows. Agents in the economy must decide on how to allocate their wealth or savings between two types of projects (i.e., the portfolio decision). The first type of projects are low-return but riskless projects. The second type of projects consists of highly productive investments that suffer from idiosyncratic shocks. Their uncertain returns are imperfectly correlated across projects. The larger the number of risky projects undertaken, the greater the diversification potential and the lower the risk faced by the global economy. In order to implement a larger number of highly productive projects and further diversify portfolios and lower global risk, however, sufficient amounts of capital are required, particularly if there are indivisible projects.

This is one of the key roles of the financial system. If the financial sector is undeveloped, it will not be able to mobilize and pool resources from a sufficiently large number of diverse and numerous savers, in a cost-effective manner, and many productive undertakings will not be possible. A more efficient financial sector would be able to fund a larger number of high productivity projects, despite their riskiness, and in this way reduce the volatility of growth. Aggregate risk declines through portfolio diversification, while the lower risk encourages investors and the higher productivity of the projects enhances economic growth.<sup>3</sup>

All of these contributions assume that agents are risk-averse. The fact that agents are sufficiently risk-averse is critical for the effect of financial development on economic growth to be positive (Bencivenga and Smith, 1991). Given risk aversion, when financial markets are poorly developed, agents chose to invest most of their wealth in a safe asset and channel only the residual to high-productivity projects. Given limited portfolio

<sup>&</sup>lt;sup>3</sup> Nevertheless, some authors claim that financial development may reduce the rate of growth (Pagano, 1993; Devereux and Smith, 1994). The reason is that, in reducing risk, diversification would allow agents to reduce their precautionary savings, which may decelerate economic growth (Mirman, 1971). If the effect of the reduction in the rate of savings on growth is stronger than the effect of the investment in more productive projects, due to diversification, the rate of growth may diminish. Which effect dominates will depend on the elasticity of intertemporal substitution.

diversification, the uncertainty related to high productivity projects is high and economic growth is slow.<sup>4</sup>

These theoretical arguments have been used to explain why developed countries have growth rates less volatile than developing countries backed by most empirical work that show a negative correlation between the variability of growth rates and the growth rate (Aizenman and Marion, 1993; Ramey and Ramey, 1995; Blattman et al. 2004; Koren and Tenreyro, 2004; Aghion et al., 2004), although the variability of growth is not explicitly analyzed and formalized in most papers.

Nevertheless, some papers analyze this question in more detail. Acemoglu and Ziliboti (1997) examine the variance of productivity which could depend negatively or positively on the number of projects implemented in the economy. These authors conclude that the variance only diminishes with financial development if productivity of risky projects is high enough and the degree of indivisibility of the projects is also high. Along the same line, Greenwood and Jovanovic (1990) find that the variance of growth rates, depends positively on the rate of return of projects, the inter-temporal discount factor and the amount of funds available for investment. Again, these authors obtain the result that the higher the amount of funds available for investment, more projects will be implemented and risk will diminish, since the portfolio is better diversified.

Lastly, Aghion, Banerjee and Piketty (1999) develop a theoretical model and show that, by mobilizing savings and facilitating the creation of reserves, the financial sector allows the economy to better absorb shocks, particularly negative shocks, which increase the likelihood that investors had difficulties to get additional funding. The idea is that with undeveloped credit markets, the demand and supply of credit become more pro-cyclical, which magnifies the effects of shocks to the economy. González-Vega and Villafanilbarnegaray (2007) show, however, that the pro-cyclical behavior of credit portfolios depends on credit technology used as well as the characteristics of producers. In particular, in some developing countries, the credit portfolios of microfinance institutions have been less pro-cyclical than those of the banking sector.

There are a number of purely empirical investigations on this first theoretical strand, based on a portfolio approach. Easterly, Islam and Stiglitz (2000) discuss the importance of financial development on growth volatility. While price and wage rigidities have been advocated to explain output

<sup>&</sup>lt;sup>4</sup> While the papers based on a portfolio approach predict that less developed countries tend to invest in secure but less productive sector, Koren and Tenreyro (2004) argue that poor countries concentrate production in a few sectors but with high specific risk (agriculture), thus rejecting the *trade off* between less volatility and higher growth. These authors show, empirically, that as countries develop, they tend to move to less volatile productive activities. Concretely, they defend the thesis that developed economies tend to specialize in sector intensive in human capital, which adapt better to external shocks (Kraay and Ventura, 2001). Further, they show that the degree of development of the financial sector does not reveal a significant effect on volatility of growth.



fluctuations, the aforementioned authors defend the hypothesis that the degree of development of the financial sector determines the stability of the economy, since higher financial development, particular the development of the stock market, permits a better management of risks.

However, more access to financial markets also allows enterprises to increase financial leverage, which could imply higher risks and higher volatility. In their empirical analysis, these authors conclude that the relationship between volatility and financial development is not linear. Thus, although higher financial development could reduce volatility initially, when financial development is high, increases in financial activity could amplify the effect of shocks on the economy. This happens specially during financial crises, when the growth of the credit market is higher than in the stock market. Related to this result, Kunieda (2008) shows, using a dynamic panel data model, that the effect of financial development on volatility is concave; in early development stages output volatility is lower, with more development volatility is higher, while with a mature financial sector volatility is again lower.

The second strand of research studies the effect of the existence of information asymmetries problems and incomplete markets on volatility of product. Some examples are Bernanke and Getler, 1990; Greenwald and Stiglitz, 1993; Kiyotaki and Moore, 1997; Carlstrom and Fuerst, 1997; Edwards and Végh, 1997; Bernanke, Gertler and Gilchrist, 1999; Jaffee and Stiglitz, 2000; De Meza and Webb, 2006. Ultimately, these problems could lead to credit rationing and inefficiencies that could reduce growth and increase volatility. Also, the financial capacity is the key to determine demand behavior as it is possible that a reduced capacity could reinforce and propagate the effects of real and monetary shocks.<sup>5</sup> To this respect, Beck et al. (2006) finds some evidence that financial intermediaries could magnify monetary shocks, particularly in countries where firms have very limited access to the capital markets.

Some models have been calibrated for the US economy and reproduce the actual movements of output (Carlstrom and Fuerst, 1997; Bernanke, Gertler and Gilchrist, 1999). In turn, Denizer et al. (2000) find that Banks could have contributed to the reduction of volatility of consumption and investment. Similarly, Jalil (2009), Ceccheti et al. (2006) and Dynan et al. (2005), find evidence that financial development reduced volatility of economic growth.

The third strand of theoretical work starts with Aghion et al. (2004), who argue that due to various market imperfections and restrictions, financial markets become less effective to facilitate the absorption of aggregate shocks

<sup>&</sup>lt;sup>5</sup> Some of these papers argue that the financial system was determinant to magnify the Great Depression of 1929. In particular, the lack of confidence in the financial institutions and the insolvency of debtors were determinants of the persistence and severity of the Great Depression. In this regard, the effect of credit restrictions on aggregate demand caused by the financial collapse (Bernanke, 1983).

which leads to higher volatility of growth. Their empirical results for a panel of countries during the period 1960-2000 show that lower financial development is associated with higher exposition to shocks and higher negative effects of volatility on growth.

Aghion and Banerjee (2005), consider the same model and conclude that in closed economies, fluctuations are detonated by the interaction between credit restrictions and interest rates, while in open economies the source of instability is the interaction between the real exchange and interest rates. Farias (2007) shows that, in the case of developed countries, volatility of investment is higher with incomplete financial markets. Also, Aghion et al. (2005), show that flexible exchange rate systems can have negative effects on productivity growth and lead to higher macroeconomic volatility when the financial sector is undeveloped; with more developed financial Systems flexible exchange rates can increase productivity growth. The authors point out the usual view that flexibility of exchange rate systems in developing countries helps to absorb real shocks and to reduce volatility (Broda, 2004; Edwards and Levy-Yeyati, 2003) requires that exchange rate volatility be less than volatility of real shocks.

Empirically, Aghion et al. (2005) find, in a sample of 83 countries from 1960 to 2000, that the effect of exchange rate volatility on economic activity is small and insignificant. More recently, Aghion et al. (2009) find that Exchange rate volatility could have a significant effect on long run productivity in the case of countries with lower levels of financial development. Also, Aghion and Marinescu (2006) argue that counter-cyclical fiscal policies have positive effects on productivity growth, particularly in countries with low degrees of financial development.

Similarly, some papers have shown the effects of liberalization and integration of the financial sector on growth and volatility, depending on the degree of financial development. Federici and Caprioli (2009) find that a high degree of financial development is critical for the existence of transmission effects among countries, following credit crises. Özbilgin (2010) shows, with a standard real business cycle model for an open economy, that financial development and market integration are associated with higher volatility of investment and output. Mallick (2009) finds that the long run variance of real GDP is affected by the degree of financial development.<sup>6</sup>

<sup>&</sup>lt;sup>6</sup> A different but related question deals with the effects of liberalization on volatility of economic growth. The first works (Obstfeld, 1994; Devereux and Smith, 1994) defend openness to external markets permits a better diversification of risk and a reduction in volatility. Nonetheless, more recent works show that the previous effect could be ambiguous. The reason is that a higher degree of international opennes makes economies less vulnerable but exposes them to external shocks (Easterly, Islam and Stiglitz, 2000). Buch, Döpke and Pierzdioch (2002) show how the effect of government policies can depend on the degree of openness of credit markets. The effect of monetary policies could be magnified if the markets are open; however, the impact of fiscal policies is reduced with more integrated financial markets. Epaular and Pommeret (2004) show that higher market integration increases volatility of income and consumption, due to more variability of foreign prices. Empirically, Kose, Prasad and Terrones (2003) conclude that the effect of financial integration on volatility is ambiguous.

Finally, it is worth mentioning the paper by Aysan (2006) that focuses on the effect of volatility on financial markets and the effect of these on economic growth, using an overlapping generations model. The author finds that higher volatility increases the costs associated with financial market imperfections, which induces higher interest rates charged by financial intermediaries and higher costs of loans; which in turn induces enterprises not to choose the more productive technologies (because they become more expensive) which leads to lower economic growth.

## 1.3 Banking sector, growth and volatility

The more recent literature points out that the way in which financial markets evolve (their structure and type of development) is also important to explain growth and volatility. Using a panel of 70 countries from 1956 to 1998 and controlling for fixed effects, Denizer et al. (2000) show that those countries with a more developed financial sector have less fluctuations in real product, consumption and investment. They also show that the share of banks in the financial system is the more robust explanation of consumption and investment volatilities, while credit offered to the private sector explains volatility of output and consumption. Although the mechanisms through which the development of the financial sector affects volatility are not clearly specified, the previous results suggest that risk management and the generation of information are particularly important to reduce volatility of consumption and investment, while the availability of credit to the private sector helps to smooth consumption and production.

Dehejia et al. (2007) argue that the institutional mechanisms that impulse financial development are important to determine its consequences on economic growth and volatility. The United States from 1900 to 1940 provides and ideal study case since during that period State banks experimented various regulations.<sup>7</sup> The financial expansion induced by the expansion of bank branching contributed to an accelerated process of mechanization of agriculture and stimulated growth of the manufacturing sector. However, the financial expansion induced by State deposit insurance had negative consequences for both sectors, since deposit insurance not only reduced the cost of loans and increased credit but also created moral hazard problems by reducing the costs of riskier credits. In fact, the State banks granted risky loans to the farming sector during the boom, previous to 1920. However, when agricultural prices collapsed during 1919 and 1920, the loans were not sustainable and this sector entered into a crises, which was rapidly propagated to the manufacturing sector.

<sup>&</sup>lt;sup>7</sup> In particular, the Glass-Steagall Act of 1933 which will be commented in the next section.

Mitchener et al. (2010) examine the relationship between the structure of the banking system and economic growth using data on the manufacturing industry and concentration of the banking sector during 1899 and 1929. This period was characterized by a rapid expansion of the manufacturing sector and the establishment of laws that restricted interstate banking, which resulted in a geographic segmentation of the banking sector and promoted concentration. These authors find that the high concentration of the banking industry affected growth of the manufacturing sector positively. A higher number of bank branches and per capita banks increased growth of those industries that depend on external financing or had more access to external sources of financing. Deposit insurance, however, depressed growth the manufacturing sector.

The previous results contrast with those in the recent literature on the effects of banking structure on economic growth, where most works show that a concentrated banking sector depresses economic growth (Cetorelli and Gambera, 2001; Clarke, 2004; Freeman, 2002; Garret et al. 2007); while more competition in this sector boosts economic growth (Claessens and Laeven 2005).<sup>8</sup>

The arguments are that higher concentration reduces credit supply and increases costs affecting growth negatively. However, the contradiction is only apparent. In general, there is agreement that concentration of banking affects positively the average growth of the economy; but different industries are affected differently depending on their requirements of external sources of funding.

Beyond the structure of the banking system, Beck et al. (2006) show that instability of macroeconomic policies increases growth volatility in countries with limited access to stock markets.

# 2. The Financial Systems in Mexico and the United States

This section presents a brief description of the financial systems in Mexico and the U. S., including an overview of their history, which was determinant of their current structure. The literature on financial development and growth attempts to explain how the financial institutions, by reducing transaction and information costs through their different functions, can affect the ultimate determinants of economic growth (investment in human and physical capital and technological change) and reduce poverty and inequality across nations

<sup>&</sup>lt;sup>8</sup> It is worth mentioning Cetorelli and Gambera (2001), who analyze data of 35 industries in 41 countries, with different degrees of economic development. These authors as well as most of the Studies in this literature agree in general that higher concentration affects positively to industries that heavily depend on external sources of funding.

(Levine, 1997). Thus, understanding the differences in the financial systems could help to understand differences in growth performance.

Most of the literature on the history of financial development points out that the enormous differences on the financial systems across countries are related to the political institutions developed in each country. The starting point is that financial development is the result of specific laws and regulatory policies in a context of conflicting interests. On the one hand, the growth of financial institutions is not possible without a government that is able to establish property rights and enforcement of financial contracts. On the other hand, governments rely on financial institutions in order to meet their financing requirements.

Thus, in the absence of institutions that could regulate governments, they will have incentives to develop financial systems that favor the flow of financial services to governments rather than to the private sector. In other words, governments can have a great influence on the development of financial systems, with consequences on economic growth. In what follows it will be seen the extent to which the previous point of view explains the enormous differences of the financial systems, particularly the banking systems, between Mexico and the US. Concretely, those differences were created by institutions that limited the power of politicians and governments.

Approximately, two centuries ago, the financial system in the US was very similar to that of Mexico. It was, essentially, a monopolized industry that generated rents for the government. However, the growing competition among the States produced, in a few decades, the emergence of hundreds of financial intermediaries all over the country. Each State had the power to grant licenses for the creation of new banks, which increased competition and reduced any tendency to monopolization. Some attempts to stop competition failed because they were not consistent with the growing competitive nature of the American political system defined by a federal system, separation of powers, electoral vote and competition among political parties.

The US decentralized system created competition among different jurisdictions and levels of government. In turn, politicians, trying to maximize profits in the financial system, created more competition in the system.<sup>9</sup> The net effect was a sort of political and financial equilibrium. This structure, together with the leadership of private agents, leaded to the development of the actual US financial system.

As a result of competition, in the actual US financial system coexist different types of financial intermediaries, including banking and non-banking institutions; and the participation of the securities markets and investment

<sup>&</sup>lt;sup>9</sup> It should be mentioned, though, that the evolution of the financial system in the US had some ups and downs. The first and second Central banks were abandoned until they disappear. The State governors incremented licenses for new banks to obtain easy funding thus avoiding application of unpopular policies and favoring political partners with profitable monopolistic activities.

funds is, nowadays, among the highest in the world. The penetration of the system (measured by the ratio of Credit to GDP) is also among the highest in the world. Further, the presence of banks at the local level is very high; actually, there are more than ten thousand institutions among commercial banks and credit unions.

The banking system has been regionally fractional since the MacFadden Act in 1927 and became specialized under the Glass-Steagall Act in 1933. The MacFadden Act prohibited banking services beyond State boundaries. The goal was to incentive competition by putting on the same ground small State banks and important national banks and allowing them to operate in a single State. As a result, the US banking industry consisted of a large number of relatively small banks.

The Glass-Steagall Act did not allow banks to offer commercial banking services, investments and insurance under the same roof. This Act was created in response to the Stock crash in 1929 and the great depression, events that lead to the crash of nine thousand banks (about a third of the existing banks in the US), representing an effort to re-establish stability and confidence on the banking system.

With the regionalized system most banks had a clear regional demarcation and with the specialization there was a strict separation among commercial banking, investment banking and insurance. However the de-regulation initiated by the end of the 1990's relaxed the previous restrictions. In particular, the Gramm-Leach Bliley Act (1999), also known as the Financial Services Modernization Act, abolished the Glass-Steagal Act and opened the doors to competition among banks, investment companies and insurance companies, allowing them to offer different services under the same roof and without restrictions.

Currently, after a sustained growth of about six years, a period of rapid consolidations, growth of bank branching and increased investment in technology, the US financial industry is stagnated as a result of the mortgage crisis that produced the collapse of leading firms and whose consequences have extended to all sectors of the economy. Pushed by growing losses, the financial institutions are concentrated in reducing costs and improving operative efficiency.<sup>10</sup> The dominant tendencies of the actual US financial system are consolidation, increased bank branching, small banking approach, strict requirements for risk management, and intermediary elimination.

On the other hand, the Mexican financial system has been characterized through history by persistent financial concentration and very limited financial inclusion. About two centuries ago, the dictator Porfirio Diaz was able to reduce and eliminate the power of State governors, making sure that congressmen candidates were their own supporters thus eliminating or

<sup>&</sup>lt;sup>10</sup> It is important to mention that, as it has always happened historically, the US financial system continues with the practice of rapidly adopting the technological advances.

limiting the power of local governments. As a consequence, the power of States to grant licenses for the creation of banks, the root of competition, was eliminated. The result was a very concentrated banking system that served the needs of the political and financial elites, repressing the development of industry in Mexico.

Despite the relatively large demand for financial services, the financial system was very small. Only a reduced group of enterprises had access to virtually unlimited funds, while the rest of economic agents faced a permanent situation of credit scarcity. The lack of institution building policies and effective protection to property rights gave rise to a financial system in Mexico, characterized by a reduced number of financial institutions that granted credits among themselves;<sup>11</sup> exactly opposite to the US system.

Differently than the US, Mexico did not develop a political structure and the institutions needed to regulate and control the power of government in the financial sector. The governments that followed Porfirio Diaz, continued exerting a tight control of the financial sector and it was not until the end of the 1990's when Mexico started to promote a more open financial sector.

The financial system in Mexico has been fundamentally a banking system, with a stock market relatively small. Until 1975 Mexico had a specialized system, somehow in the same spirit of the US Glass-Steagall Act. Later on, however, the system became more oriented to universal banking, creating financial groups with some restrictions, similar to the ones of the Glass-Steagal Act. However, differently that in the US, in Mexico there was nothing similar to the MacFadden Act, that could have imposed regional restrictions.

During most of the twentieth century this sector has suffered various financial repressions.<sup>12</sup> Besides, the sector was expropriated by government in 1982, re-privatized in 1990, crashed in 1995 and, since 1998, had increased participation of global banks. Despite the measures taken during the financial liberalization by the end of the 1980's and early 1990's, the Mexican banks have reduced loans in recent years because of liquidity constraints, legal restrictions that difficult the repayment of debts, higher risks and the consequent no-payment culture.

The penetration of the banking system in Mexico, of about 17% on average, is relatively low when compared with various similar economies. Despite its technology, which has international standards, the banking sector is small. There are approximately 40 banks, out of which only 5 concentrate more than 80% of total assets. On the other hand, 65% of the labor force does not have access to financial services of commercial banks; that is, at least 25 million

<sup>&</sup>lt;sup>11</sup> By the end of the XIX century, for example, existed two banks, with networks at the national level, which controlled more than 60% of total assets.

<sup>&</sup>lt;sup>12</sup> These repressions have been diverse and included the orientation of credit to specific sectors, interest rate ceilings, credit rationing, subsidies, absence of property rights and negative real interest rates during inflationary periods.

people out of the total labor force, do not have formal financial services. The formal financial sector, constituted by banks, stock market and other intermediaries, only reaches about 35% of people in the labor force.<sup>13</sup>

According to the Financial Culture Survey,<sup>14</sup> only 40% of the Mexican population owns a bank account, 69% does not have information on the availability of financial products and services, and 8 out of 10 Mexicans prefer to have cash in hand rather than a credit or debit card. The total savings operated by formal financial intermediaries represent less than 60% of GDP, while the total formal credit represents only 15% of GDP. Thus, the formal financial sector in Mexico is not well developed and a relatively small fraction of the population uses formal financial services.

There have been some attempts to expand the financial sector by promoting the expansion of small financial entities, such as credit and savings unions and other microfinance institutions. However, the so called semiformal financial sector, has not grown as expected and has had a limited penetration in the economy, basically through cooperatives that offer some financial services to the poor in rural areas. The development of financial services in rural areas can be characterized by having excessive costs, a large number of condonance programs, excessive regulation and a regressive effect on income distribution. As a result, government transfers to the rural sector diminished during the 1990's.

In the past, the Mexican cooperatives did not have legal support or formal mechanisms to promote their development under certainty. However, in June 2001, Congress passed on the new Law of Popular Savings and Credit (LACP) and the statutory Law of the National Bank of Financial Services (BANSEFI). The LACP establishes a monitoring system for the Popular Savings and Credit Sector through the National Commission of Banking and Stocks (CNBV) and BANSEFI, both of which are government institutions. This law consolidated the popular savings and credit sector by generating scale economies and facilitating the channeling of transfers from government programs through these institutions.

According to popular savings and credit institutions, an important problem faced by them is the great difficulty to operate under the legal framework. In this sense, the microfinance institutions have to look for sources of funding in order to assure their functioning and to be authorized by second level institutions, federations and the CNBV, thus sacrificing their operative sutainability. Another problem, from the point of view of the microfinance institutions is the lack of coordination among the existing government

<sup>&</sup>lt;sup>13</sup> See "La Banca Popular como un instrumento para incorporar a los beneficios de la globalización a quienes se encuentran ajenos a éstos". Series of Conferences at Universidad La Salle, "Globalización en el Siglo XXI". México, DF. April, 2002.

<sup>&</sup>lt;sup>14</sup> The survey was carried out by the National Bank of Mexico (BANAMEX) and the Autonomous National University of Mexico (UNAM), in 2008.

programs. As a result, since the 1980's several informal mechanisms of savings and credit have been developed for the people excluded from the formal financial sector but they also have been used by non excluded population. These mechanisms include a variety of financial transactions that do not have any regulation.

As a conclussion, in can be inferred from the previous discussion that the US financial sector is characterized by a high degree of development and penetration as well as a high level of competition along history, despite of the concentration of State banks in some periods. However, in the case of Mexico, the formal financial system, despite the support of public policies has not been able to reach most of the population and the informal financial sector has been more successful. Nowadays, high concentration and financial exclussion of a large sector of the population still presist, as in most developing countries.

# 3. Empirical Time Series Approach

Differently than most empirical papers in the related literature, this paper focuses on two specific countries, Mexico and the US using time series models. The empirical approach includes the following aspects. First, it proceeds to characterize the dynamics of the relevant variables. Next, it models output growth and financial development as a VAR process and performs causality tests and impulse-response analysis. Finally, a relevant GARCH model is estimated in order to examine the relationship between financial development and growth volatility. In what follows each of these aspects is briefly described.

# 3.1. Characterization of the variables

First of all, it will be determined if the dynamics of the growth rates of output and measures of financial development, is consistent with stationary processes by applying unit root tests. This aspect is fundamental in order to avoid potentially misleading inferences. Four unit root tests will be used, namely, the augmented Dickey-Fuller (Dickey and Fuller, 1979, 1981), Dickey-Fuller GLS (Elliot, Rothenberg and Stock, 1996), MS-t (Ng and Perron, 2001) and the KPSS (Kwiatkowsky, Phillips, Schmidt and Shin, 1992) tests. As it is well known, the null hypothesis for the first three tests is that the process has a unit root, while the last test considers stationarity as the null hypothesis. Since the empirical analysis focuses on growth rates, the tests will include a constant only in the estimated test equations in all cases.

## 3.2. VAR model, causality and impulse-response functions

A controversial aspect in the empirical literature on financial development and growth is on the direction of this relationship. In this paper, neither output growth nor financial development is assumed exogenous but rather it is assumed that there is a dynamic inter-relationship between the two processes. For this reason, the following VAR model is specified:

$$\begin{bmatrix} y_t \\ x_t \end{bmatrix} = \begin{bmatrix} \alpha_{11} \\ \alpha_{22} \end{bmatrix} + \sum_{j=1}^p \begin{bmatrix} \beta_{11,j} & \beta_{12,j} \\ \beta_{21,j} & \beta_{22,j} \end{bmatrix} \begin{bmatrix} y_{t-j} \\ x_{t-j} \end{bmatrix} + \begin{bmatrix} u_{yt} \\ u_{xt} \end{bmatrix}$$
(1)

Where y represents the growth rate of real GDP and x is a measure of financial development.<sup>15</sup> The coefficients  $\beta_{12,j}$  capture the effect of financial development on growth, while the coefficients  $\beta_{21,j}$  indicate the opposite effect, that from growth to financial development. The parameters  $\alpha_{11}$  and  $\alpha_{22}$  can be interpreted as the autonomous components of both processes respectively. The terms  $u_{yt}$  and  $u_{xt}$  are random shocks which satisfy the conventional assumptions of zero mean, constant variance and constant contemporaneous covariance. The sub-index  $j = 1, \dots, p$  indicates the lag number and its maximum p will be determined using optimal lag criteria and assuring that the estimated residuals do not exhibit any significant autocorrelation pattern. In this way, we will avoid potential miss specification problems.

Once an appropriate VAR model has been established, the interdependence between both variables will be evaluated through causality tests in the sense defined by Granger (1969) and Sims (1972). Specifically, it will be evaluated if the coefficients  $\beta_{12,j}$ ;  $j = 1, \dots, p$  are jointly equal to zero, in which case financial development will not affect (or Ganger-cause) growth. On the other hand, if the coefficients  $\beta_{21,j}$ ;  $j = 1, \dots, p$  are jointly equal to zero, output growth will not affect (or Granger-cause) financial development.

Lastly, impulse-response functions of growth of real GDP to orthogonal shocks to financial development will be presented. It is important to mention that, in both cases, test for possible seasonality effects will be carried out, and in the cases where they are found significant, will be controlled for by including seasonal dummies.<sup>16</sup> Also, in the case of Mexico, the model will

CIDE

<sup>&</sup>lt;sup>15</sup> As we will see later, these are the ratios of Domestic Credit and Money Supply (M2), both in nominal terms, to nominal GDP.

<sup>&</sup>lt;sup>16</sup> This aspect is particularly important since we will use quarterly data in both cases.

include the growth rate of US's real GDP as an exogenous variable in order to capture the well known link between the US and Mexican economies.

#### 3.3. Time series model of growth and volatility

In order to evaluate the dynamics of growth and its volatility the following univariate time series model will be considered:<sup>17</sup>

$$y_{t} = \alpha_{11} + \sum_{j=1}^{m} \beta_{11,j} y_{t-j} + \sum_{j=1}^{n} \beta_{12,j} x_{t-j} + \phi \sigma_{t} + u_{yt}$$
(2)

$$\sigma_t^2 = \mu + \gamma \sigma_{t-1}^2 + \lambda u_{t-1}^2 + \psi y_{t-1} + \varphi x_{t-1}$$
(3)

Where equation (2) is the conditional mean of the growth process and is, essentially, the first equation of the VAR model described in previous subsection, augmented by a GARCH-in-mean effect. The coefficients  $\beta_{12,j}$  capture the effect of financial development on growth.

The parameter  $\phi$  measures the extent to which volatility of growth affects the average growth rate. As mentioned before, although most empirical papers find a negative correlation between volatility and growth, from a theoretical point of view this relationship could be positive or negative. On the one hand, higher volatility reduces the risk-adjusted return of investment which has a negative effect on growth; on the other, higher volatility of growth increases growth by increasing precautionary savings and investment. Another possible mechanism is worth mentioning since this study uses short run data. A positive value of the parameter  $\phi$  would also be consistent with Black's (1987) hypothesis, according to which investment in more specialized and risky technologies could lead to higher but more volatile growth rates.

The values of m and n are the number of lags for the growth and financial development processes respectively and they may not be equal nor do they necessarily refer to consecutive sequences of lags.

Equation (3) models the conditional variance of  $u_{yt}$  as a GARCH (1, 1) process.<sup>18</sup> This process will characterize the dynamics of growth volatility.<sup>19</sup>

<sup>&</sup>lt;sup>17</sup> The justification for using a single equation approach is that, as we will see in the next section, the measures of financial development used in this study, despite their variability, do not show ARCH effects. Preliminarily, we proceeded to characterize the dynamics of real GDP growth as well as the measures of financial development and to evaluate if they exhibit a volatility pattern through LM tests for ARCH effects. ARCH effects were found only for the growth of real GDP processes but not for the financial development measures. The exception was the ratio Domestic Credit to GDP for the US case.

 $<sup>^{18}</sup>$  It makes justice to mention that these models initiate with the pioneering work by Engle (1982) and Bollerslev (1986).

<sup>&</sup>lt;sup>19</sup> This specification will turn out to be appropriate for the case of the US. For the Mexican case, volatility of growth will be modeled as an ARCH (1) process only.

The parameter  $\varphi$  will measure the effect of financial development on volatility of real GDP growth. Based on the literature previously discussed, in general, a more developed financial sector could reduce growth volatility since economic agents would have more information and better capacity to diversify productive risks and to deal with liquidity shocks. The parameter  $\psi$  will indicate, if positive, that higher growth rates in the economy will become more volatile, implying a feedback effect of growth on its own volatility.

In the case of Mexico, the growth rate of real GDP of the US economy will be included as a possible determinant of volatility of Mexican growth. It should be remarked that this is a novel issue investigated in this study. The fact that the Mexican economy is closely related to the US economy in terms of economic activity and other important macroeconomic variables, such as interest and exchange rates, makes it reasonable to postulate, empirically, that the growth rate of the US economy could also affect volatility of growth in Mexico.

Econometrically, the most appropriate GARCH model will be determined as follows. First, the appropriate specification for the conditional mean equation will be found by considering a maximum value of 24 for m and n, and selecting those lags that are statistically significant. Then, the specification is considered satisfactory if the estimated residuals do not show any significant patterns of autocorrelation. With this specification at hand, the possible presence of conditional heteroskedasticity will be evaluated, using the LM test for ARCH effects. After finding evidence on these effects, the ARCH(1) and GARCH (1, 1) specifications will be considered the more plausible specifications since the study uses quarterly data and, therefore, it is not expected to find too much persistence in the variance processes.

The model given by equations (2) and (3) is qualitatively different to the one used in some growth regressions with cross-country data, where growth volatility is approximated by the variance or standard deviation of growth rates of a group of economies at each period of time. Although this measure of volatility is accepted in the empirical literature for practical reasons, it is certainly limited to represent the inherent uncertainty or volatility of the growth process over time. This can be illustrated with the following example. Suppose two economies with highly volatile but identical growth rates. If we consider the standard deviation of this group of two economies as a measure of volatility, we would conclude that there is no volatility since the standard deviation of growth rates but very different to each other. In this case, the standard deviation will indicate that there is (high) volatility when in fact there is none.

The time series approach used in this study will allow us to get around the previous problem, since it makes possible to measure the variance of the process at each period of time, conditional on all previous information, to

evaluate if this process changes over time; and, if this is the case, to select the appropriate ARCH/GARCH specification. In addition it will be possible to model volatility effects on the mean of the process by allowing for GARCH-inmean effects as well as by considering possible determinants of the volatility process itself, as it will be done in the present study. As it was mentioned before, this study concentrates on two economies only, thus avoiding the need of imposing *a priori* restrictions on equality of coefficients across economies.

The approach, however, could have some limitations. First, it uses a bivariate model which could be subject to the critique of under specification. More importantly, this model is not derived from a structural dynamic model, which would allow more proper evaluation of shocks to the economy. A related shortcoming is that the econometric results could be consistent with different mechanisms. Secondly, since the study uses quarterly data, the model may capture mainly short run effects, which may differ from long run effects, more related to the growth process. Finally, the time span of the study is not very large, particularly, in the case of Mexico, which can produce potentially biased and inefficient results. For all these reasons, this study should be considered a first approximation and the results should be taken with caution.

## 4. Empirical Results

This section first describes the variables used in the study as well as the data sources; then, the main findings are presented and discussed, following the empirical methodology outlined in the previous section.

## 4.1. Data sources and variables

This study uses quarterly data obtained from the International Monetary Fund's *International Financial Statistics* CD's. The data is available for the period 1957:01-2009:04 for the United States and for the period 1981:01-2009:03 for Mexico. The primary variables are Nominal Domestic Credit, Money Supply (M2), Nominal Gross Domestic Product and GDP Implicit Deflator.

With the previous variables the growth rate of real GDP (GREALGDP) has been constructed and will be taken here as an indicator of economic growth. Also, two indicators of financial development have been obtained. One of them is defined as the growth rate of the ratio of Nominal Domestic Credit to nominal GDP (GDCRGDP). The other indicator of financial development is the growth rate of the ratio of Nominal Money Supply to nominal GDP (GM2GDP). In the related empirical literature these indicators are considered measures of credit and monetary deepening respectively and are accepted measures of financial development.

## 4.2 Unit root tests

Table A-1, shown in the Appendix, shows the unit root testing results. For the case of the US all tests show that GREALGDP and GM2GDP are consistent with stationary processes. For the variable GDCRGDP all tests but the ADF indicate stationarity. For practical reasons this process will be considered stationary.

For the Mexican case, there is strong evidence to conclude that GM2GDP is stationary. For the variables GREALGDP and GDCRGDP there is one test in each case that indicates that the process has a unit root. However, in all cases the KPSS test indicates that the variables are stationary. Although the evidence is not as strong as it would have been desirable, we consider that it is sufficient to characterize all three variables (expressed in annualized growth rates) as stationary.<sup>20</sup>

# 4.3 VAR model, Granger-causality tests and impulse-response functions

Table 1 below reports the results on causality tests. These tests were performed using estimation results of the VAR models reported in table A-2, shown in the Appendix. It is important to remark that these specifications were considered satisfactory alter making sure that the estimated residuals did not show any significant autocorrelation patterns.

As it can be seen, in the case of US the results support mutual causality between growth of the ratio domestic credit to GDP (GDCRGDP), that is growth of credit deepening, and growth of real GDP (GREALGDP). Regarding the relationship between growth of the ratio M2 to GDP (GM2RGDP), that is growth of monetary deepening, and growth of real GDP there is evidence on unidirectional causality only, from GREALGDP to GM2RGDP. Thus, in the case of US, the growth of real GDP could have been affected by credit deepening but not by monetary deepening.

<sup>&</sup>lt;sup>20</sup> Based on purely economic considerations, it would be very hard to justify, for example, that the growth rate of real GDP could have a unit root. Besides, the data does not seem to exhibit such a behavior. Looking at the autocorrelation and partial autocorrelation functions, the apparent unit root behavior of the series (in growth rates) seems to be a consequence of seasonality and big fluctuations observed in the data.



NULL HYPOTHESIS	VAR	SEASONAL	TEST	<i>P-</i> VALUE
	ORDER	DUMMIES	(D. O. F)	
UNITED STATES (1957:01 – 2009:0	04)			
GDCRGDP does not Granger-cause grealgdp	4	Yes	24.52 (4)	0.0001***
GREALGDP does not Granger- cause gdcrgdp	4	Yes	16.64 (4)	0.0023***
GM2GDP does not Granger-cause GREALGDP	12	No	14.65 (12)	0.2609
GREALGDP does not Granger-cause GM2GDP	12	No	24.27 (12)	0.0187**
MÉXICO (1981:01 – 2009:03)				
GDCRGDP does not Granger-cause GREALGDP	11	Yes	16.58 (11)	0.1210
GREALGDP does not Granger-cause GDCRGDP	11	Yes	53.72 (11)	0.0000***
GM2GDP does not Granger-cause GREALGDP	6	Yes	11.74 (6)	0.0679*
GREALGDP does not Granger-cause GM2GDP	6	Yes	21.40 (6)	0.0016***

TABLE 1. GRANGER-CAUSALITY TESTS

In the case of the US 205 observations were used in the first two tests and 191 in the other two. In the case of Mexico the two first tests used 102 observations while the other two tests used Only 89 observations. Numbers in parenthesis in the fourth column (d. o. f.) indicate the degrees of freedom of the Chi-square distribution of the corresponding test.

In the case of Mexico, in contrast, the evidence suggests that credit deepening does not seem to affect growth of real GDP; on the contrary, growth of real GDP seems to have affected growth of credit deepening. It should be remarked that this study uses total domestic credit to measure financial development. Considering credit to the private sector could lead to different results.

Also, it is important to consider that, in the case of Mexico, the public debt represents an important fraction of the total bank's assets compared to credit to the private sector; and that an important part of the financing of private investment is done through foreign banks. Further work, that takes into account the previous distinctions, could shed more light about the role of growth of credit supply on economic growth.

As far as the second indicator of financial development, namely monetary deepness, approximated here by the growth of the ratio of money supply (M2) to GDP (GM2GDP), there is evidence of a bi-directional relationship between this indicator and growth of real GDP. However, it should be pointed out that the relationship from GM2GDP to output growth is statistically weak, while the opposite relationship (reverse causality) is statistically strong.

In order to have a better understanding of the previous relationships we next turn to impulse-response analysis. Figure 1 shows the impulse-response functions of real GDP growth to orthogonal shocks of one standard deviation on financial development measures. While in the case of US positive shocks to both indicators of financial development seem to affect positively the growth of real GDP for as long as one year; in the case of Mexico, a positive shock to growth of credit deepening does not seem to affect real GDP growth and a positive shock to growth of monetary deepening seems to have a positive effect on output growth, but it lasts for one quarter only.

Thus, in a country such a Mexico where monetary deepening seems to be particularly low, improvements on the monetization of the economy could have positive effects on the growth rate of real GDP, although these effects would not be persistent. A possible explanation for the fact that credit deepening does not seem to affect output growth could be a crowding-out effect resulting from the allocation of credit to the public sector.

## 4.4 Financial development and growth volatility

Table 2 below presents estimation results of the model described in section 3.3, equations (2) and (3), for the cases of US and Mexico. It is important to make some considerations. First, the time spans are not the same in both cases. Approximately, the number of observations for the case of US doubles that of Mexico. Thus, in the Mexican case, the econometric results could be subject to some biases and inefficiencies.



FIGURE 1. IMPULSE-RESPONSES OF GROWTH TO SHOCKS ON FINANCIAL DEVELOPMENT

These are impulse-response functions to one orthogonal innovation to measures of financial development. The results are based on the VAR estimation results shown in the appendix, Table A-2.

Second, in the case of Mexico we observed seasonality and, therefore, seasonal dummies were included in the estimation. Third, in the case of Mexico, the growth of US's real GDP is included explicitly as a possible determinant of volatility, a novel aspect of this paper. Finally, while in the case of the US it was found a conditional heteroskedasticity pattern, consistent with a GARCH (1, 1) process, in the case of Mexico an ARCH (1) was sufficient to capture the dynamics of conditional variance.<sup>21</sup>

<sup>&</sup>lt;sup>21</sup> Following the methodology described in the previous section, in all cases the residuals of the final specification chosen for the conditional mean do not exhibit any correlation patterns. Nevertheless, there is evidence of ARCH effects. The results are not presented here but they are available upon request.

	MODEL 1:		MODELO 2:		
	GROWTH OF REAL GDP		GROWTH OF REAL GDP		
	GROWTH OF	D. CREDIT/GDP	GROWTH OF M2/PIB		
	UNITED STATES	MEXICO	UNITED STATES	MEXICO	
		(CONDITIO	NAL MEAN)		
Constant	1.734 (0.0274)	-8.768 (0.0154)	2.861 (0.0030)	-1.730 (0.6626)	
Lags of Y	0.061 (0.0650)	-0.042 (0.0086)	0.075 (0.0376)	-0.025 (0.0662)	
Lags of X	-0.029 (0.0503)	0.009 (0.0735)	0.032 (0.5523)	n. a.	
$\phi$	0.186 (0.4761)	0.433 (0.2117)	-0.249 (0.3678)	-0.688 (0.0514)	
Growth of US	n. a.	1.215 (0.0001)	n. a.	0.719 (0.0405)	
		(Conditiona	al Variance)		
Constant	0.619 (0.2458)	27.995 (0.0144)	1.549 (0.0409)	50.148 (0.0000)	
ARCH (1)	0.169 (0.0360)	0.668 (0.082)	0.231 (0.0259)	0.263 (0.0261)	
GARCH(1)	0.806 (0.0000)	n. a.	0.728 (0.0000)	n. a.	
ψ	-0.122 (0.3593)	0.001 (0.9986)	-0.292 (0.1365)	-0.362 (0.2638)	
φ	0.094 (0.2010)	0.012 (0.9606)	0.083 (0.5244)	-1.137 (0.0010)	
Growth of US	n. a.	-3.949 (0.0457)	n. a.	-3.366 (0.1336)	
Adjusted $R^2$	0.19	0.83	0.14	0.81	
Log-likelihood	-449.05	-277.49	-445.18	-306.61	
<i>F</i> -test ( <i>p</i> -value)	4.18 (0.0000)	22.43 (0.0000)	3.17 (0.0002)	32.23 (0.0000)	
Seasonal Dummies	No	Yes	No	Yes	
# of obs.	186	92	182	94	

TABLE 2. GARCH-M ESTIMATION RESULTS

All models were estimated by maximum likelihood using the numerical optimization algorithm Marquardt, with the software EViews 6.0. Numbers in parenthesis are p-values. To save space, rows "Lags of Y" and "Lags of X" we report the average coefficients on all lags included in the conditional mean equation, as well as their corresponding average *p*-values.

In all cases, after estimating the full model the standardized residuals and their squares were examined, finding no evidence of autocorrelation and

therefore the estimated models could be considered well specified. In what follows, the most relevant aspects of the estimation results are discussed.

In the first place, Mexican growth is directly related to US growth, a fact that is very well known. It is important to point out that this pulling effect that the US economy has on the Mexican economy, is quite robust and appears virtually in any specification considered.

Secondly, for the case of US, the parameter  $\phi$  is not statistically significant, which could imply that the effects of volatility on investment returns and precautionary savings cancel out. Alternatively, Black's (1987) hypothesis that higher volatility will be positively related to average growth rates of the economy is not confirmed. However, in the case of Mexico, the parameter  $\phi$  was found significant (model 2) but negative, which contradicts the hypothesis of Black, pointing more to the interpretation that the negative effect of volatility on return to investment dominates, so that higher volatility implies lower average growth rates in the Mexican economy.

Third, although volatility of growth is much less persistent in Mexico than in the US, in terms of level it is much higher in Mexico than in the US. Taking into account the estimated parameters of the conditional variance processes, the unconditional variances of US growth rates are approximately 24.8 and 37.8, in models 1 and 2 respectively; in the case of Mexico these magnitudes are 84.3 and 68.0 respectively.

Fourth, in all cases the parameter  $\psi$  was not statistically significant implying that the average growth rates of real GDP do not affect volatility of growth in the subsequent periods. Thus, we find no evidence of feedback effects from output growth to volatility.

Fifth, regarding the effect of financial development on volatility of growth of real GDP, captured by the parameter  $\varphi$ , while in the case of US there is no evidence on this effect, in the case of Mexico we find a negative and statistically significant value for this parameter in the case of model 2. Thus, we find some evidence that financial development (measured with the indicator of monetary deepening) reduces volatility of growth in the case of Mexico.

Interestingly, the findings  $\phi < 0$  and  $\phi < 0$  taken together imply a virtuous circle between financial development and volatility of growth: higher financial development reduces volatility of growth and, given that  $\phi$  is negative, lower volatility of growth would produce a higher growth rate of real GDP, this last effect being consistent with the hypothesis that less volatile growth will produce higher investment rates, which would lead to higher growth rates of output.

Finally, the last row of the conditional variance results suggest that the growth rate of US's real GDP could also have a benign effect on Mexican growth through volatility. In both models, US growth has a negative effect on

volatility of Mexican growth and in Model 1 this effect was found significant at the 5% (in Model 2 the *p*-value is of 0.13 and, therefore, the estimated effect could be considered marginally significant).

Thus, higher growth in the US not only increases Mexican growth directly, a fact that is very well known, but also by reducing its volatility, which is negatively related to Mexico's real GDP growth. Certainly, in the opposite case, a reduction in US's growth will lead to lower growth, an effect that could be amplified by the subsequent increased in volatility.

# Conclusions

This paper has investigated empirically, using time series methods, the effect of financial development on growth of real GDP and its volatility in the cases of Mexico and the United States. The theoretical literature suggests that financial development could affect not only the average growth rate of real output but also its volatility, although the literature on this last link is not vast. In general, a better functioning of the financial sector could reduce volatility of growth by promoting better diversification of productive risks, improving the capacity to face liquidity shocks, mobilizing savings, mainly precautionary, and allocating capital to more efficient investments. The paper also explores the possible effect of US's output growth on volatility of Mexican growth, a channel that is worth investigating given the enormous influence of the US economy on Mexico's economic performance.

The Granger-causality results suggest that both in the US and Mexico there is a bi-directional relationship between real GDP growth and one measure of financial development in each case, although the evidence is not quite strong. Certainly, further work is necessary to clarify this issue, perhaps in the context of structural models.

Regarding the effects of finance on growth and volatility this study has found that, in the case of the US, financial development is related to output growth but does not affect growth volatility. On the other hand, growth volatility does not seem to affect the average growth rate of this economy. In the case of Mexico, however, it was found that growth of the financial sector is associated with lower levels of growth volatility, and that there is a negative relationship between growth volatility of growth and the average growth rates of real output.

Finally, this study has also found that higher growth in the US not only affects positively Mexico's average growth rate, a fact that is very well known, but also it reduces growth volatility which, given the aforementioned negative relationship between growth volatility and average growth, will increases the average growth rate of this economy. The study thus explicitly finds a new channel from US growth to Mexican growth by reducing Mexican growth volatility.

# References

- Acemoglu, D. and F. Zilibotti (1997), "Was Prometheus Unbound by Chance? Risk, Diversification and Growth", *Journal of Political Economy*, 105, 709-751.
- Acemoglu, D., P. Aghion and F. Zilibotti (2002), "Distance to Frontier, Selection, and Economic Growth", National Bureau of Economic Research Working Paper N° 9066.
- Aghion, P. and I. Marinescu (2006), "Cyclical Budgetary Policy and Economic Growth: What Do We Learn from OECD Panel Data?", Working Paper.
- Aghion, P. and A. Banerjee (2005), *Volatility and Growth*, Clarendon Lectures, Oxford University Press.
- Aghion, P. and P. Howitt (1992), "A Model of Growth through Creative Destruction," *Econometrica*, 60, 2 (March), 323-351.
- Aghion, P., A. Banerjee and T. Piketty (1999), "Dualism and Macroeconomic Volatility", *Quarterly Journal of Economics*, 114, 1359-1397.
- Aghion, P., M. Angeletos, A. Banerjee and K. Manova (2004), "Volatibility and Growth: The Role of Financial Development", Harvard University (Department of Economics), mimeo.
- Aghion, P., P. Bacchetta, R. Ranciere and K. Rogoff (2005), "Productivity Growth and the Exchange Rate Regime: The Role of Financial Development", Harvard University, mimeo.

(2006), "Exchange Rate Volatility and Productivity Growth: The Role of Financial Development", *Journal of Monetary Economics*, 56, 4, 494-513.

- Aghion, P., T. Fally and S. Scarpeta (2006), "Credit Constraints as a Barrier to the Entry and Post-Entry Growth of Firms: Lessons from Firm Level Cross Country Panel Data." Working Paper, Harvard University (October).
- Ahlin, C. and N. Jiang (2005, "Can Microcredit Bring Development". Working Paper No. 05-019. Department of Economics. Nashville: Vanderbilt University.
- Aizenman, J. and N. Marion (1993), "Policy Uncertainty, Persistence and Growth", *Review of International Economics*, 2, 145-163.
- Ang, J. B. and W. J. McKibbin (2007), "Financial liberalization, financial sector development and growth: Evidence from Malaysia", *Journal of Development Economics*, 84, 215-233.
- Arestis, P., P. O. Demetriades, and K. B. Luintel (2001), "Financial Development and Economic Growth: The Role of Stock Markets", *Journal of Money, Credit, and Banking*, 33, 16-41.

- Aysan, A. F. (2006), "The Effects of Volatility on Growth and Financial Development Through Capital Market Imperfections", Munich Personal RePEc Archive (MPRA) Paper, No. 5486.
- Bagehot, W. (1873), *Lombard Street: A Description of the Money Market*. Philadelphia: Orion Editions.
- Beck, T., D. Asli and V. Maksimovic (2006), "The Influence of Financial and Legal Institutions on Firm Size", *Journal of Banking and Finance*, 30, 2995-3015.
- Beck, T., R. Levine and N. Loayza (2000), "Finance and the Sources of Growth", Journal of Financial Economics, 58, 261-300.
- Bencivenga, V. R. and B.D. Smith (1991), "Financial Intermediation and Endogenous Growth". *Review of Economic Studies*, 58, 195-209.

\_\_\_\_\_ (1993). "Some Consequences of Credit Rationing in an Endogenous Growth Model", *Journal of Economic Dynamics and Control*, 17, 97-122.

- Bernanke, B. (1983), "Non-Monetary Effects of the Financial Crisis in the Propagation of the Great Depression", *American Economic Review*, 73, 257-276.
- Bernanke, B., M. Gertler and S. Gilchrist (1999), "The Financial Accelerator in a Quantitative Business Cycle Framework", en J. B. Taylor and M. Woodford, eds, *Handbook of Macroeconomics*, Vol. 1 C, cap. 2. North Holland.
- Bernanke, B. and M. Gertler (1989), "Agency Costs, Net Worth, and Business Fluctuations," *American Economic Review*, 79, 14-31.
- Black, F. (1987), "Bussines Cycles and Equilibrium, New York: Basil Blackwell.
- Blattman, C., J. Hwang and J. Williamson (2004), "The Impact of the Terms of Trade on Economic Development in the Periphery, 1870-1939: Volatility and Secular Change," NBER Working Paper No. 10600.
- Bollerslev, T. (1986), "Generalized Autoregressive Conditional Heteroskedasticity." *Journal of Econometrics*, 31, 307-327.
- Broda, C. (2004), "Terms of Trade and Exchange Rate Regimes in Developing Countries", *Journal of International Economics*, 63, 31-58.
- Buch C.M., J. Döpke and C. Pierzdioch (2002), "Financial Openness and Business Cycle Volatility," Kiel Working Paper No. 1121, Kiel Institute of World Economics.
- Buiter, W. H. and K. M. Kletzer (1995), "Capital Mobility, Fiscal Policy, and Growth under Self-financing of Human Capital Formation", *Canadian Journal of Economics*, 28, S163-S194.
- Cameron, R. et al. (1967). Banking in the Early Stages of Industrialization: A Study in Comparative Economic History. New York: Oxford University Press.

- Carlstrom, C.T. and T. S. Fuerst (1997), "Agency Costs, Net Worth, and Business Fluctuations: A Computable General Equilibrium Analysis", *American Economic Review*, 87, 5, 893-910.
- Cecchetti, S. G., A. Flores-Lagune and S. Krause (2006), "Financial Development, Consumption Smoothing, and the Reduced Volatility of Real Growth", AEA Conference Papers.
- Cetorelli, N. and M. Gambera (2001), "Banking Market Structure, Financial Dependence, and Growth: International Evidence from Industry Data", *Journal of Finance*, 106(2), 617-48.
- Claessens, S. and L. Laeven (2005), "Financial Dependence, Banking Sector Competition, and Economic Growth", *Journal of the European Economic Association*, 3(1), 179-207.
- Clarke, M. Z. (2004), "Geographic Deregulation of Banking and Economic Growth", *Journal of Money, Credit, and Banking*, 36(5), 929-942.
- De Gregorio, J. (1996), "Borrowing Constraints, Human Capital Accumulation, and Growth", *Journal of Monetary Economics*, 37, 49-72.
- De Gregorio, J. and S. Kim (2000), "Credit Markets with Differences in Abilities: Education, Distribution and Growth", *International Economic Review*, 41, 3, 579-607.
- De la Fuente, A. and J. M. Marin (1996), "Innovation, Bank Monitoring and Endogenous Financial Development", *Journal of Monetary Economics*, 38, 269-301.
- De Meza, D. and D. Webb (2006), "Credit Rationing: Something's Gotta Give." *Economica*, 73, 563-578.
- Dehejia, R. and A. Lleras-Muney (2007), "Financial Development and Pathways of Growth: State Branching and Deposit Insurance Laws in the United States 1900-1940", *The Journal of Law and Economics*, 50.
- Demetriades, P. and K. Hussein (1996), "Does Financial Development Cause Economic Growth? Time Series Evidence from 16 Countries", *Journal of Development Economics*, 51, 387-411.
- Denizer, C., M.F. lyigun and A.L. Owen (2000), "Finance and Macroeconomic Volatility", Policy Research Working Papers 2487, The World Bank.
- Devereux, M. B. and G. W. Smith, G. W. (1994), "International Risk Sharing and Economic Growth", *International Economic Review*, 35, 535-550.
- Dickey, D. and W. Fuller (1979), "Distribution of the Estimators for Autoregresive Time Series with a Unit Root." *Journal of the American Statistical Association*, 74, 427-431.
- Dickey, D. and W. Fuller (1981), "Likelihood Ratio Tests for Autoregresive Time Series with a Unit Root." *Econometrica*, 49, 1057-1072.

- Dynan, K. E., D. W. Elmendorf and D. E. Sichel (2005), "Can financial innovation explain the reduced volatility of economic activity?", Carnegie-Rochester Conference on Public Policy.
- Easterly, W., R. Islam and J. Stiglitz (2000), "Shaken and Stirred: Explaining Growth Volatility", *World Bank Economic Review*.
- Edwards, S. and C.A. Végh (1997), "Banks and Macroeconomics Disturbances under Predetermined Exchange Rates", *Journal of Monetary Economics*, 40, 239-278.
- Edwards, S. and E. Levy-Yeyati (2003), "Flexible Exchange Rates as Shock Absorbers", National Bureau of Economic ResearchWorking Papers No. 9867.
- Elliot, G., Rothenberg, T. and J. Stock (1996), "Efficient Tests for an Autoregressive Unit Root", *Econometrica*, 64, 813-36.
- Engle, R. (1982), "Autoregressive Conditional Heteroskedasticity with Estimates of the Volatility of UK Inflation", *Econometrica*, 50, 987-1007.
- Epaulard, A. and Pommeret, A. (2005), "Financial Integration, Growth, and Volatility", IMF Working Papers 05/67, International Monetary Fund.
- Farías, M. E. (2007), "Uncertainty and Volatility on Financial Markets: The Effects on Growth and Development", Universidad de Chile.
- Federici, D. and F. Caprioli (2009), "Financial Development and Growth: An Empirical Analysis", *Economic Modelling*, 26(2), 285-294.
- Freeman, D. G. (2002), "Did State Bank Branching Deregulation Produce Large Growth Effects?", *Economics Letters*, 75(3), 383-389.
- Fry, M. (1988), *Money, Interest and Banking in Economic Development*. Baltimore: The Johns Hopkins University Press.
- Fung, M. (2009), "Financial Development and Economic Growth: Convergence or Divergence?", *Journal of International Money and Finance*, 28, 56-67.
- Galbis, V. (1977), "Financial Intermediation and Economic Growth in Less-Developed Countries: A Theoretical Approach." Journal of Development Studies, 13(2), 58-72.
- Galor, O. and J. Zeira (1993), "Income Distribution and Macroeconomics", *Review of Economic Studies*, 60, 35-52.
- Garrett, T. A., G. A. Wagner and D. C. Wheelock (2007), "Regional Disparities in the Spatial Correlation of State Income Growth, 1977-2002", *Annals of Regional Science*, 41(3), 601-18.
- Giuliano, P. and M. Ruiz-Arranz (2009), "Remittances, Financial Development and Growth", *Journal of Development Economics*, 90, 1, 144-152.
- Goldsmith, R.W. (1969), *Financial Structure and Development*. New Haven, CT: Yale University Press.

- González-Vega, C. and M. Villafani-Ibarnegaray (2007), "Las Microfinanzas en la Profundización del Sistema Financiero. El Caso de Bolivia." *Trimestre Económico*, No.293, 5-68.
- Granger, C. (1969). "Investigating Causal Relationships by Econometric Models and Cross Spectral Methods." *Econometrica*, 37, 424-438.
- Greenwald, B. and J. Stiglitz (1993), "Financial Market Imperfections and Business Cycles", *Quarterly Journal of Economics*, 108, 77-114.
- Greenwood, J. and B. Jovanovic (1990), "Financial Development, Growth, and the Distribution of Income", *Journal of Political Economy*, 98, 1076-1107.
- Greenwood, J., J. M. Sanchez and C. Wang (2010), "Quantifying the Impact of Financial Development on Economic Development", Working Paper 10-05, Federal Reserve Bank of Richmond.
- Grossman, G. M. and E. Helpman (1991), *Innovation and Growth in the Global Economy*. Cambridge MA: MIT Press.
- Gup, E.B., ed. (2003), The Future of Banking, Westport, CT.: Quorum.
- Gurley, J. G. and E. S. Shaw (1955). "Financial Aspects of Economic Development." *American Economic Review*, 45(4), 515-538.

\_\_\_\_\_ (1960). *Money in a Theory of Finance*. Washington, D.C.: The Brookings Institution.

- Hamilton, A. (1791), "National Bank", in *The Works of Alexander Hamilton*, ed. By H. C. Lodge, Volume 3, New York: G. P. Putman's Sons, 1904.
- Hasan, I., P. Wachtel and M. Zhou (2009), "Institutional development, financial deepening and economic growth: Evidence from China", *Journal of Banking & Finance*, 33(1), 157-170.
- Hicks, J. (1969), A Theory of Economic History. Oxford: Clarendon Press.
- Jacoby, H. G. and E. Skoufias (1997), "Risk, Financial Markets, and Human Capital in a Developing Country", *The Review of Economic Studies*, 64, 311-335.
- Jaffee D. and J. Stiglitz (2000), "Credit Rationing" en el *Handbook of Monetary Economics*, Elsevier.
- Jalil, A. (2009), "Financial Development and Reduced Growth Volatility: Evidence from China", International Trade and Finance Association Conference Papers, 19th International Conference Working Papers, 21.
- Jones, L., R. Manuelli and E. Stacchetti (2000), "Technology and Policy Shocks in Models of Endogenous Growth," Federal Reserve Bank of Minneapolis Working Paper 281.
- Jung, W. S. (1986), "Financial Development and Economic Growth: International Evidence", *Economic Development and Cultural Change*, 34, 333-346.
- Kapur, B. K. (1976), "Alternative Stabilization Policies for LessDeveloped Economies." *Journal of Political Economy*, 84 (4), 777-795.

- King, R. G. and R. Levine (1993), "Finance, Entrepreneurship and Growth: Theory and Evidence", *Journal of Monetary Economics*, 32, 513-542.
- Kiyotaki, N. and J. Moore (1997), "Credit Cycles", *Journal of Political Economy*, 105, 2, 211-248.
- Koren, M. and Tenreyro, S. (2004), "Diversification and Development," Harvard University and Federal Reserve Bank of Boston, mimeo.
- Kose, M. A., E.S. Prasad E.S. and M.E. Terrones (2003), "Financial Integration and Macroeconomic Volatility", IMF Working Paper 03/50 (Washington: International Monetary Fund).
- Kower, P. (2002), "A Model of Schumpeterean Growth with Institutional Change and Financial Development: Theory and Evidence". Doctoral Dissertation. Columbus, Ohio: The Ohio State University (Available at: http://aede.osu.edu/programs/RuralFinance/publications.htm).
- Kraay, A. and J. Ventura (2001), "Comparative Advantage and the Crosssection of Business Cycles", NBER working paper 8104.
- Kunieda, T. (2008), "Financial Development and Volatility of Growth Rates: New Evidence", Ryukoku University, mimeo.
- Kwiatkowsky, P., Schmidt and Shin (1992), "Testing the Null Hypothesis of Stationary against the Alternative of a Unit Root." *Journal of Econometrics*, 54, 159-178.
- La Porta, R., F. Lopez-de-Silanes, A. Shleifer and R. Vishny (1999), "Corporate Ownership around the World", *Journal of Finance*, 54, 471-517.
- Levine, R. (1991), "Stock Markets, Growth, and Tax Policy", *Journal of Finance*, 46, 1445-1465.
- Levine, R. (1992), "<u>Financial Structures and Economic Development</u>", Policy Research. Working Paper Series 849, The World Bank.

\_\_\_\_\_ (1997), "Financial Development and Economic Growth: Views and Agenda," *Journal of Economic Literature*, 35, 688-726.

- \_\_\_\_\_ (2004), "Finance and Growth: Theory and Evidence", NBER Working Paper Series, 10766.
- Levine, R. and S. Zervos (1998), "Stock Markets, Banks and Economic Growth", *American Economic Review*, 88, 357-558.
- Levine, R., L. A. Laeven and S. Michalopoulos (2009), "Financial Innovation and Endogenous Growth", CEPR Discussion Paper No. DP7465 (September).
- Levine, R., N. Loayza and T. Beck (2000), "Financial Intermediation and Growth: Causality and Causes", *Journal of Monetary Economics*, 46: 31-77.
- Loayza, N. and R. Ranciere (2002), "Financial Fragility, Financial Development, and Growth", World Bank, mimeo.
- Lucas, R. E. Jr. (1988), "On the Mechanics of Development Planning", Journal of Monetary Economics, 22, 1 (July), 3-42.

Mallick, D. (2009), "Financial Development, Shocks, and Growth Volatility", MPRA Paper 17799, University Library of Munich, Germany.

Masten, A. B., F. Coricelli and I. Masten (2008), "Non-Linear Growth Effects of Financial Development: Does Financial Integration Matter?", *Journal* of International Money and Finance, 27(2), 295-313.

Mathieson, D. J. (1980). "Financial Reform and Stabilization Policy in a Developing Economy." *Journal of Development Economics*, 7 (3), 359-395.

McKinnon, R.I. (1973). *Money and Capital in Economic Development*. Washington, D.C: The Brookings Institutions.

, ed. (1976). Money and Finance in Economic Growth and Development: Essays in Honor of Edward S. Shaw. New York: Marcel Dekker.

Merton, R. C. (1992), "Financial Innovation and Economic Performance", Journal of Applied Corporate Finance, 4, 12-22.

- Mirman, L. (1971), "Uncertainty and Optimal Consumption Decisions", *Econometrica*, 39, 179-85.
- Mitchener, K. J. and D. C. Wheelock (2010), "Does the Structure of Banking Markets Affect Economic Growth? Evidence from U.S. State Banking Markets", NBER working paper 15710.
- Morales, M.F. (2003), "Financial Intermediation in a Model of Growth through Creative Destruction", *Macroeconomics Dynamics*, 7, 363-393.
- Ng, S. and P. Perron (2001), "LAG Length Selection and the Construction of Unit Root Tests with Good Size and Power", *Econometrica*, 69, 1519-1554.

Obstfeld, M. (1994), "Risk-Taking, Global Diversification, and Growth", *American Economic Review*, 84, 1310-1329.

Özbilgin, H. M. (2010), "Financial market participation and the developing country business cycle", *Journal of Development Economics*, 92(2), 125-137.

Pagano, M. (1993), "Financial Markets and Growth: An Overview", *European Economic Review*, 37, 613-622.

Rajan, R. G. and L. Zingales (1998), "Financial Dependence and Growth", *American Economic Review*, 88, 559-586.

Ramey, G. and V. Ramey (1995), "Cross-Country Evidence on the Link between Volatility and Growth", *The American Economic Review*, 85, 1138-1151.

Robinson, J. (1952), "The Generalization of the General Theory" en *The Rate of Interest and other essays.* London: Macmillan, pp. 67-142.

Romer, P. (1986), "Increasing Returns and Long-Run Growth", Journal of Political Economy, 94, 5, (October), 1002-1037.

Romer, P. (1990), "Endogenous Technological Change", *Journal of Political Economy*, 98, 5 (October), part II, S71-S102.

Saint-Paul, G. (1992), "Technological Choice, Financial Markets and Economic Development", *European Economic Review*, 36, 763-781.

- Schumpeter, J. A. (1934). The Theory of Economic Development. Cambridge, MA: Harvard University Press.
- Shaw, E. S. (1973). *Financial Deepening in Economic Development*. New York: Oxford University Press.
- Sims, C. (1972). "Money, Income and Causality." *American Economic Review*, 62, 540-552.

#### DIVISIÓN DE ADMINISTRACIÓN PÚBLICA

- Salvador Espinosa, On Bond Market Development and Strategic Cross-Border Infrastructure..., DTAP-269.
- Ignacio Lozano, Ejidos y comunidades: ¿cuarto nivel de gobierno?..., DTAP-268.
- Ernesto Flores y Judith Mariscal, *Oportunidades y desafíos de la banda ancha móvil en América Latina*, DTAP-267.
- Judith Mariscal y Walter Lepore, *Oportunidades y uso de las TIC: Innovaciones en el Programa de combate a la pobreza*, DTAP-266.
- Ernesto Flores y Judith Mariscal, *El caso de la Licitación de la Red Troncal en México: Lecciones para el Perú*, DTAP-265.
- Dolores Luna *et al.*, *Índice de Gobierno Electrónico Estatal: La medición 2010*, DTAP-264.
- Gabriel Purón Cid y J. Ramón Gil-García, *Los efectos de las características tecnológicas en los sitios web del gobierno*, DTAP-263.
- Ana Elena Fierro y J. Ramón Gil-García, *Más allá del acceso a la información*, DTAP-262.
- Gabriel Purón Cid, Resultados del "Cuestionario sobre la reforma Presupuesto basado en Resultados...", DTAP-261.
- Guillermo Cejudo y Alejandra Ríos, *El acceso a la información gubernamental en América Central y México: Diagnóstico y propuestas*, DTAP-260.

#### División de Economía

- Kurt Unger, *Especializaciones reveladas y condiciones de competitividad en las entidades federativas de México*, DTE-530.
- Antonio Jiménez, *Consensus in Communication Networks under Bayesian Updating*, DTE-529.
- Alejandro López, *Environmental Dependence of Mexican Rural Households*, DTE-528. Alejandro López, *Deforestación en México: Un análisis preliminar*, DTE-527.
- Eva Arceo, Drug-Related Violence and Forced Migration from Mexico to the United States, DTE-526.
- Brasil Acosta *et al.*, *Evaluación de los resultados de la Licitación del Espectro Radioeléctrico de la COFETEL*, DTE-525.
- Eva Arceo-Gómez and Raymundo M. Campos-Vázquez, ¿Quiénes son los NiNis en México?, DTE-524.
- Juan Rosellón, Wolf-Peter Schill and Jonas Egerer, *Regulated Expansion of Electricity Transmission Networks*, DTE-523.
- Juan Rosellón and Erix Ruíz, *Transmission Investment in the Peruvian Electricity Market: Theory and Applications*, DTE-522.
- Sonia Di Giannatale *et al.*, *Risk Aversion and the Pareto Frontier of a Dynamic Principal-Agent Model: An Evolutionary Approximation*, DTE-521.

#### **DIVISIÓN DE ESTUDIOS INTERNACIONALES**

Mariana Magaldi and Sylvia Maxfield, *Banking Sector Resilience and the Global Financial Crisis: Mexico in Cross-National Perspective*, DTE-229.

Brian J. Phillips, Explaining Terrorist Group Cooperation and Competition, DTE-228.

- Covadonga Meseguer and Gerardo Maldonado, *Kind Resistance: Attitudes toward Immigrants in Mexico and Brazil*, DTEI-227.
- Guadalupe González *et al.*, *The Americas and the World 2010-2011. Public Opinion and Foreign Policy in Brazil, Colombia, Ecuador, Mexico and Peru*, DTEI-226.
- Guadalupe González *et al., Las Américas y el mundo 2010-2011: Opinión pública y política exterior en Brasil, Colombia, Ecuador, México y Perú,* DTEI-225.
- Álvaro Morcillo Laiz, Un vocabulario para la modernidad. Economía y sociedad de Max Weber (1944) y la sociología en español, DTEI-224.
- Álvaro Morcillo Laiz, Aviso a los navegantes. La traducción al español de Economía y sociedad de Max Weber, DTEI-223.
- Gerardo Maldonado, *Cambio electoral, anclaje del voto e intermediación política en sistemas de partidos de baja institucionalización*, DTEI-222.
- James Ron and Emilie Hafner-Burton, *The Latin Bias: Regions, the Western Media and Human Rights*, DTEI-221.
- Rafael Velázquez, La política exterior de Estados Unidos hacia México bajo la administración de Barack Obama, DTEI-220.

#### DIVISIÓN DE ESTUDIOS JURÍDICOS

- Rodrigo Meneses y Miguel Quintana, Los motivos para matar: Homicidios instrumentales y expresivos en la ciudad de México, DTEJ-58.
- Ana Laura Magaloni, *La Suprema Corte y el obsoleto sistema de jurisprudencia constitucional*, DTEJ-57.
- María Mercedes Albornoz, *Cooperación interamericana en materia de restitución de menores*, DTEJ-56.
- Marcelo Bergman, Crimen y desempleo en México: ¿Una correlación espuria?, DTEJ-55.
- Jimena Moreno, Xiao Recio y Cynthia Michel, *La conservación del acuario del mundo. Alternativas y recomendaciones para el Golfo de California*, DTEJ-54.
- María Solange Maqueo, *Mecanismos de tutela de los derechos de los beneficiarios*, DTEJ-53.
- Rodolfo Sarsfield, *The Mordida's Game. How institutions incentive corruption*, DTEJ-52.
- Ángela Guerrero, Alejandro Madrazo, José Cruz y Tania Ramírez, *Identificación de las estrategias de la industria tabacalera en México*, DTEJ-51.
- Estefanía Vela, Current Abortion Regulation in Mexico, DTEJ-50.
- Adriana García and Alejandro Tello, *Salaries, Appelate Jurisdiction and Judges Performance*, DTEJ-49.

#### DIVISIÓN DE ESTUDIOS POLÍTICOS

- Gilles Serra, The Risk of Partyarchy and Democratic Backsliding: Mexico's Electoral Reform, DTEP-238.
- Allyson Benton, *Some Facts and Fictions about Violence and Politics in Mexico*, DTEP-237.
- Allyson Benton, *The Catholic Church, Political Institutions and Electoral Outcomes in Oaxaca, Mexico*, DTEP-236.
- Carlos Elizondo, *Stuck in the Mud: The Politics of Constitutional Reform in the Oil Sector in Mexico*, DTEP-235.
- Joy Langston and Francisco Javier Aparicio, *Gender Quotas are not Enough: How Background Experience and Campaigning Affect Electoral Outcomes*, DTEP-234.
- Gilles Serra, How Could Pemex be Reformed? An Analytical Framework Based on Congressional Politics, DTEP-233.
- Ana Carolina Garriga, *Regulatory Lags, Liberalization, and Vulnerability to Systemic Banking Crises*, DTEP-232.
- Rosario Aguilar, *The Tones of Democratic Challenges: Skin Color and Race in Mexico*, DTEP-231.
- Rosario Aguilar, Social and Political Consequences of Stereotypes Related to Racial Phenotypes in Mexico, DTEP-230.
- Raúl C. González and Caitlin Milazzo, An Argument for the 'Best Loser' Principle in Mexico, DTEP-229.

#### **DIVISIÓN DE HISTORIA**

- Michael Sauter, Spanning the Poles: Spatial Thought and the 'Global' Backdrop to our Globalized World, 1450-1850, DTH-77.
- Adriana Luna, La reforma a la legislación penal en el siglo XVIII: Notas sobre el aporte de Cesare Beccaria y Gaetano Filangieri, DTH-76.
- Michael Sauter, Human Space: The Rise of Euclidism and the Construction of an Early-Modern World, 1400-1800, DTH-75.
- Michael Sauter, Strangers to the World: Astronomy and the Birth of Anthropology in the Eighteenth Century, DTH-74.
- Jean Meyer, Una revista curial antisemita en el siglo XIX: Civiltá Cattolica, DTH-73.
- Jean Meyer, Dos siglos, dos naciones: México y Francia, 1810-2010, DTH-72.
- Adriana Luna, La era legislativa en Nápoles: De soberanías y tradiciones, DTH-71.
- Adriana Luna, *El surgimiento de la Escuela de Economía Política Napolitana*, DTH-70.
- Pablo Mijangos, *La historiografía jurídica mexicana durante los últimos veinte años*, DTH-69.
- Sergio Visacovsky, "Hasta la próxima crisis". Historia cíclica, virtudes genealógicas y la identidad de clase media entre los afectados por la debacle financiera en la Argentina (2001-2002), DTH-68.

#### ESTUDIOS INTERDISCIPLINARIOS

- Ugo Pipitone, México y América Latina en la tercera oleada (crecimiento, instituciones y desigualdad), DTEIN-02.
- Eugenio Anguiano, El estudio de China desde cuatro enfoques: histórico, político, internacionalista y económico, DTEIN-01.

El CIDE es una institución de educación superior especializada particularmente en las disciplinas de Economía, Administración Pública, Estudios Internacionales, Estudios Políticos, Historia y Estudios Jurídicos. El Centro publica, como producto del ejercicio intelectual de sus investigadores, libros, documentos de trabajo, y cuatro revistas especializadas: *Gestión y Política Pública, Política y Gobierno, Economía Mexicana Nueva Época* e *Istor*.

Para adquirir cualquiera de estas publicaciones, le ofrecemos las siguientes opciones:

VENTAS DIRECTAS:	VENTAS EN LÍNEA:
Tel. Directo: 5081-4003 Tel: 5727-9800 Ext. 6094 y 6091 Fax: 5727 9800 Ext. 6314	Librería virtual: www.e-cide.com
Av. Constituyentes 1046, 1er piso, Col. Lomas Altas, Del. Álvaro Obregón, 11950, México, D.F.	Dudas y comentarios: publicaciones@cide.edu

# ¡¡Colecciones completas!!

Adquiere los CDs de las colecciones completas de los documentos de trabajo de todas las divisiones académicas del CIDE: Economía, Administración Pública, Estudios Internacionales, Estudios Políticos, Historia y Estudios Jurídicos.



¡Nuevo! ¡¡Arma tu CD!!

PUBLICACION	IS ODE
	(0.0)
	Arma to CD

Visita nuestra Librería Virtual <u>www.e-cide.com</u> y selecciona entre 10 y 20 documentos de trabajo. A partir de tu lista te enviaremos un CD con los documentos que elegiste.